Author's response to reviews

Title: A hypothyroid child with rhabdomyolysis and recent history of cardiac tamponade: a case report

Authors:

Juan J Delgado (juandelgado@ufm.edu)
Waleska Guevara (waleskaguevara@gmail.com)
Evelyn Ramos (doresly@hotmail.com)
Claudia Lorenzana (claudialorenzana17@yahoo.com)
Susana Soto (susy2874@yahoo.com)

Version: 3 Date: 6 July 2011

Author's response to reviews:

The modifications I made:

1. Added the postal address of every author
2. Included the ethnicity of the patient
3. Modified the consent and specified it was obtained from the "patient’s legal guardian"
4. Removed ALL text from the figures